import { TypeList } from "docs-ui"

# sales-channel

## Interfaces

- [BaseFilterable](../../sales_channel/interfaces/sales_channel.BaseFilterable/page.mdx)
- [Context](../../sales_channel/interfaces/sales_channel.Context/page.mdx)
- [CreateSalesChannelDTO](../../sales_channel/interfaces/sales_channel.CreateSalesChannelDTO/page.mdx)
- [FilterableSalesChannelProps](../../sales_channel/interfaces/sales_channel.FilterableSalesChannelProps/page.mdx)
- [FindConfig](../../sales_channel/interfaces/sales_channel.FindConfig/page.mdx)
- [IMessageAggregator](../../sales_channel/interfaces/sales_channel.IMessageAggregator/page.mdx)
- [IModuleService](../../sales_channel/interfaces/sales_channel.IModuleService/page.mdx)
- [ISalesChannelModuleService](../../sales_channel/interfaces/sales_channel.ISalesChannelModuleService/page.mdx)
- [JoinerServiceConfig](../../sales_channel/interfaces/sales_channel.JoinerServiceConfig/page.mdx)
- [JoinerServiceConfigAlias](../../sales_channel/interfaces/sales_channel.JoinerServiceConfigAlias/page.mdx)
- [MessageAggregatorFormat](../../sales_channel/interfaces/sales_channel.MessageAggregatorFormat/page.mdx)
- [RestoreReturn](../../sales_channel/interfaces/sales_channel.RestoreReturn/page.mdx)
- [SalesChannelDTO](../../sales_channel/interfaces/sales_channel.SalesChannelDTO/page.mdx)
- [SalesChannelLocationDTO](../../sales_channel/interfaces/sales_channel.SalesChannelLocationDTO/page.mdx)
- [SoftDeleteReturn](../../sales_channel/interfaces/sales_channel.SoftDeleteReturn/page.mdx)
- [UpdateSalesChannelDTO](../../sales_channel/interfaces/sales_channel.UpdateSalesChannelDTO/page.mdx)
- [UpsertSalesChannelDTO](../../sales_channel/interfaces/sales_channel.UpsertSalesChannelDTO/page.mdx)

## Type Aliases

- [Event](../../sales_channel/types/sales_channel.Event/page.mdx)
- [EventMetadata](../../sales_channel/types/sales_channel.EventMetadata/page.mdx)
- [FindConfigOrder](../../sales_channel/types/sales_channel.FindConfigOrder/page.mdx)
- [JoinerRelationship](../../sales_channel/types/sales_channel.JoinerRelationship/page.mdx)
- [LinkModulesExtraFields](../../sales_channel/types/sales_channel.LinkModulesExtraFields/page.mdx)
- [Message](../../sales_channel/types/sales_channel.Message/page.mdx)
- [MetadataType](../../sales_channel/types/sales_channel.MetadataType/page.mdx)
- [ModuleJoinerConfig](../../sales_channel/types/sales_channel.ModuleJoinerConfig/page.mdx)
- [ModuleJoinerRelationship](../../sales_channel/types/sales_channel.ModuleJoinerRelationship/page.mdx)
- [RawMessageFormat](../../sales_channel/types/sales_channel.RawMessageFormat/page.mdx)
